Author’s response to reviews

Title: Reoperation Following Vitrectomy for Diabetic Vitreous Hemorrhage With versus Without Preoperative Intravitreal Bevacizumab

Authors:
Xuting Hu (sbeard78@163.com)
qintuo pan (panqintuo@qq.com)
jingwei zheng (jingwei-66@163.com)
Zongming Song (szmeyes@126.com)
Zongduan Zhang (zzduan@163.com)

Version: 5 Date: 19 Jul 2019

Author’s response to reviews:

Zongduan Zhang MD
Eye Hospital of Wenzhou Medical University,
270 Xueyuan Road,
Wenzhou, Zhejiang 325027, China.
Fax: (86)-577-88068831.
Email: zzduan@163.com.

July 20, 2019

Dear Editor,

We much appreciate the editor’s and reviewer’s positive and constructive comments and suggestions regarding our manuscript entitled “Reoperation Following Vitrectomy for Diabetic Vitreous Hemorrhage With versus Without Preoperative Intravitreal Bevacizumab” (BOPH-D-18-00514R2).

We have studied the editor’s comments carefully and have made revisions according to the comments.

Point-by-Point Response
1. Please represent authors’ names using their initials, not their full name, in the Authors’ Contributions section of your Declarations. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors. For example Xuting Hu should be 'XH' in this section.
- Please upload your revised manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. Please also ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.

Reply: We are so sorry that we misunderstand the meaning of ‘the authors’ names using their initials’, and thank you for your example. We have represented the authors' names using their full initials in the Authors’ Contributions section on page 14. Thank you for your suggestion. We have uploaded a manuscript as a single, final, and clean version.

We tried our best to improve writing and made some changes in the manuscript. These changes will not influence the content and framework of the paper. We appreciate for Editors’ and Reviewers’ warm work earnestly and hope that the correction will meet with approval.

We appreciate your consideration of our manuscript, and we look forward to receiving comments from the reviewers.
Thank you again and best regards.

Sincerely,

Zongduan Zhang